Author's response to reviews

Title: Atypical presentation of Renal Cell Carcinoma: A case report

Authors:

Deepak D Doshi Mr (drdeepdoshi@yahoo.com)
Michael M Saab Mr (michael.saab@pat.nhs.uk)
Singh N Singh Dr (drnids@rediffmail.com)

Version: 4 Date: 4 March 2007

Author's response to reviews:

We have received an email from you that the letter written above should be incorporated in the manuscript. We shall write this at the beginning of the manuscript and re-submit the manuscript. This will explain to the reviewers, the importance of the case, the lessons learnt from the case report and how it differs from the previous cases in the literature.

Thanks

D Doshi et al